Author’s response to reviews

Title: Improving implementation of evidence-based practice in mental health service delivery: The role of staff-focused values interventions

Authors:

Virginia Williams Ms (vcw961@uow.edu.au)
Lindsay G Oades Dr (loades@uow.edu.au)
Frank P Deane Prof (fdeane@uow.edu.au)
Trevor P Crowe Dr (tcrowe@uow.edu.au)
Joseph V Ciarrochi Prof (ciarrochij@gmail.com)
Retta Andresen Dr (retta@uow.edu.au)

Version: 2 Date: 11 May 2013

Author’s response to reviews:

The revisions requested have been addressed within the manuscript. Please refer to the cover letter attached.